Author's response to reviews

Title: Association of the single nucleotide polymorphism in chromosome 9p21 and chromosome 9q33 with coronary artery disease in Chinese population

Authors:

Qi Li (16261967@qq.com)
Wenhui Peng (pwenhui@tongji.edu.cn)
Hailing Li (lihailin_2004@126.com)
Jianhui Zhuang (piscesjhz@sina.com)
Xuesheng Luo (lxsy2009@sina.com)
Yawei Xu (xuyawei@tongji.edu.cn)

Version: 4 Date: 09 Sep 2017

Author’s response to reviews:

Dear Editor,

According to the CrossRef version, we have revised the manuscript. However, some overlap could not be totally revised since the statement of methodology may be the same between different studies.

Thank you for your consideration.

Yawei Xu